322' Copyright Ron Burback 1997More Advanced FeaturesRon BurbackLecture 19

323' Copyright Ron Burback 1997Abbreviationsnauth: authentication, authorizationncom: communicationndce: distributed computing environmentndflt: defaultnelt: elementnep: endpointnexp: expiration

324' Copyright Ron Burback 1997Abbreviations, continuednfn: functionnid: identifiernif: interfaceninq: inquirenmbr: membernmgmt: managementnns: name service

325' Copyright Ron Burback 1997Abbreviations, continuednprotect: protectionnprotseq: protocol sequencenprinc: principalnrpc: remote procedure callnstats: statistics

326' Copyright Ron Burback 1997Managing UUIDnuuid_createnuuid_equal

327' Copyright Ron Burback 1997Finding Servicesnrpc_ns_binding_import_begin, _next, _done (client)—Get a list of binding handles for a particular servicenrpc_ns_binding_select—Get one binding

328' Copyright Ron Burback 1997Announcing Servicesnrpc_ns_binding_export—Export a binding to the name space. (server)nrpc_ns_binding_unexportnrpc_server_inq—return binding information from the runtime. (server)

329' Copyright Ron Burback 1997RPC Security Clientnrpc_binding_set_auth_info—Associates security information with a binding handle. (client)nrpc_if_register_auth_info—Associates security information with an interface. (client)

330' Copyright Ron Burback 1997RPC Security Servernrpc_binding_inq_auth_client—Returns security information associated with a client (server)nrpc_server_register_auth_info—Specifies the security options used by the server. (server)

331' Copyright Ron Burback 1997Object ID Managementnrpc_binding_inq_object—Get the UUID of this objectnrpc_object_set_type—Set the UUID of an object

332' Copyright Ron Burback 1997End Point Managementnrpc_ep_register—Provides server information to the endpoint mapper. (server)nrpc_ep_unregister

333' Copyright Ron Burback 1997Starting and Stopping a Servicenrpc_server_listennrpc_mgmt_stop_server_listening

334' Copyright Ron Burback 1997Finding Protocol Sequencesnrpc_network_inq_protseqsnrpc_network_is_protseq_validnrpc_server_use_all_protseqsnrpc_server_use_all_protseqs_ifnrpc_server_use_protseqnrpc_server_use_protseq_epnrpc_server_use_protseq_if

335' Copyright Ron Burback 1997Server Group/Profile Managementnrpc_ns_group_mgr_addnrpc_ns_group_mgr_removenrpc_ns_profile_elt_addnrpc_ns_ profile_elt_remove

336' Copyright Ron Burback 1997Interface Managementnrpc_server_register_if—register an interface with the runtime. (server)

337' Copyright Ron Burback 1997ACL Managementnsec_acl_mgr_is_authorized—Determines if the specified permissions are available to a principal.

338' Copyright Ron Burback 1997Key Managementnsec_key_mgmt_get_keynsec_key_mgmt_manage_key—periodically changes server key before it expires

339' Copyright Ron Burback 1997Loginsec_login_get_current_contextnsec_login_set_contextnsec_login_setup_identitynsec_login_validate_identity